Author’s response to reviews

Title: Improvement in Early Urinary Continence Recovery After Robotic-assisted Radical Prostatectomy Based on Postoperative Pelvic Anatomic Features: A Retrospective Review

Authors:
Akihiro Nakane (aki-nakane@ams.odn.ne.jp)
Hiroki Kubota (479659@kainan.jaaikosei.or.jp)
Yusuke Noda (yusukenoda0401@gmail.com)
Tomoki Takeda (tmk7888@gmail.com)
Yasuhiko Hirose (yyhhhh3@gmail.com)
Atsushi Okada (a-okada@med.nagoya-cu.ac.jp)
Kentaro Mizuno (kmizuno@med.nagoya-cu.ac.jp)
Noriyasu Kawai (n-kawai@med.nagoya-cu.ac.jp)
Keiichi Tozawa (toza@med.nagoya-cu.ac.jp)
Yutaro Hayashi (yutaro@med.nagoya-cu.ac.jp)
Takahiro Yasui (yasui@med.nagoya-cu.ac.jp)

Version: 1 Date: 25 Jun 2018

Author’s response to reviews:

1. In the Ethics approval and consent to participate section, if consent was not required, please provide details of why this was waived and also the name of institution/committee who approved this.

Answer: The name of institution/committee the is the Ethical Committee of the Aichi Prefectural Federation of Agricultural Cooperatives for Health and Welfare Kainan Hospital. I changed this section comment.
2. Please provide a Consent to publish section in the Declarations. As your manuscript does not contain any individual persons’ identifying data, please state “Not applicable” in this section.

Answer: I state “Not applicable” in this section.

3. On uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript.

Answer: I changed my manuscript.